Author's response to reviews

Title: Use Of Who Adolescent Job Aid Algorithms By Health Workers For Reproductive Morbidities Among Rural Adolescent Girls In North India

Authors:

Archana Siddaiah (archanapink@yahoo.com)
Baridalyne Nongkrynh (baridalyne@gmail.com)
Anand Krishnan (kanandiver@yahoo.com)
Chandrakant S Pandav (cpandav@iqplusin.org)

Version: 2 Date: 11 November 2013

Author's response to reviews: see over
COVER LETTER

To,

The Editor,

BMC Public Health

Respected Sir/Madam,

Subject: Submission of revised manuscript.

We are submitting our revised manuscript titled “Use of Who Adolescent Job Aid Algorithms by Health Workers for Reproductive Morbidities among Rural Adolescent Girls in North India”.

Changes made include-

1. Addition of authors’ contribution after the competing interests section.
2. Addition of acknowledgement section after authors’ contribution section.
3. Declaration of funding for the study followed by acknowledgement.

This we wish to submit under original research column. We also declare that this manuscript has not been submitted elsewhere for publication. There are no competing interests.

Thank you.

Authors